Author's response to reviews

Title: Prognostic significance of Hypoxia-Inducible Factor 1 alpha (HIF-1alpha) expression in advanced poorly differentiated serous ovarian cancer: an immunohistochemical study

Authors:

Alexandros Daponte (dapontea@otenet.gr)
Maria Ioannou (mioan@med.uth.gr)
Ilias Mylonis (dapontea@otenet.gr)
George Simos (poroi@otenet.gr)
Marcos Minas (markosminas@hotmail.com)
Ioannis E Messinis (messinis@med.uth.gr)
George Koukoulis (poroi@otenet.gr)

Version: 3 Date: 23 June 2008

Author's response to reviews:

Dear Editor,

As suggested in your e-mail in the Methods section of our article the sentence "Institutional Review Board (IRB) approval was obtained" will be replaced by "The explorative laparotomies, chemotherapies and the treatment of patients included in this trial was carried out in the University Hospital of Larissa - University of Thessalia Greece. All the necessary informed consents as requested by the Institutional Review Board (IRB) of the University Hospital were given in writing."

Sincerely yours

Al. Daponte MD FCOG
Corresponding author